////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints]]++org.elasticsearch.index.fielddata.ScriptDocValues.GeoPoints++::
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-arcDistance-2]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#arcDistance%2Ddouble%2Ddouble%2D[arcDistance](double, double)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-arcDistanceWithDefault-3]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#arcDistanceWithDefault%2Ddouble%2Ddouble%2Ddouble%2D[arcDistanceWithDefault](double, double, double)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-geohashDistance-1]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#geohashDistance%2Djava.lang.String%2D[geohashDistance](<<painless-api-reference-String,String>>)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-geohashDistanceWithDefault-2]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#geohashDistanceWithDefault%2Djava.lang.String%2Ddouble%2D[geohashDistanceWithDefault](<<painless-api-reference-String,String>>, double)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-get-1]]<<painless-api-reference-org-elasticsearch-common-geo-GeoPoint,org.elasticsearch.common.geo.GeoPoint>> link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#get%2Dint%2D[get](int)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getLat-0]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getLat%2D%2D[getLat]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getLats-0]]double[] link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getLats%2D%2D[getLats]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getLon-0]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getLon%2D%2D[getLon]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getLons-0]]double[] link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getLons%2D%2D[getLons]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getValue-0]]<<painless-api-reference-org-elasticsearch-common-geo-GeoPoint,org.elasticsearch.common.geo.GeoPoint>> link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getValue%2D%2D[getValue]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-getValues-0]]<<painless-api-reference-List,List>> link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#getValues%2D%2D[getValues]()++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-planeDistance-2]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#planeDistance%2Ddouble%2Ddouble%2D[planeDistance](double, double)++
* ++[[painless-api-reference-org-elasticsearch-index-fielddata-ScriptDocValues-GeoPoints-planeDistanceWithDefault-3]]double link:{elasticsearch-javadoc}/org/elasticsearch/index/fielddata/ScriptDocValues.GeoPoints.html#planeDistanceWithDefault%2Ddouble%2Ddouble%2Ddouble%2D[planeDistanceWithDefault](double, double, double)++
* Inherits methods from ++<<painless-api-reference-Collection,Collection>>++, ++<<painless-api-reference-Iterable,Iterable>>++, ++<<painless-api-reference-List,List>>++, ++<<painless-api-reference-Object,Object>>++
